Author’s response to reviews

Title: Interdisciplinary and multiprofessional outpatient secondary individual prevention of work-related skin diseases in the metalworking industry: 1-year follow-up of a patient cohort

Authors:
Annika Wilke (awilke@uos.de)
Günther Gediga (guenther@gediga.de)
Andreas Goergens (Andreas.Goergens@bghm.de)
Andreas Hansen (andreas.hansen@uni-osnabrueck.de)
Anja Hübner (anjhuebn@uni-osnabrueck.de)
Swen John (johnderm@uos.de)
Kathrin Nordheider (kathrin.nordheider@uni-osnabrueck.de)
Marc Rocholl (mrocholl@uos.de)
Sabine Weddeling (Sabine.Weddeling@klinikumdo.de)
Britta Wulfhorst (britta.wulfhorst@medicalschool-hamburg.de)
Dorothee Nashan (Dorothee.Nashan@klinikumdo.de)

Version: 2 Date: 28 Nov 2018

Author’s response to reviews:

We have uploaded a cover letter including a point-by-point response to the comments of the decision letter.